Author’s response to reviews

Title: Effects of Radix Linderae extracts on a mouse model of diabetic bladder dysfunction in later decompensated phase

Authors:

Xufeng Yang (1274780649@qq.com)
Dawei Lian (675836257@qq.com)
Pinglong Fan (1160964159@qq.com)
Yifei Xu (330851639@qq.com)
Jing Wang (591071542@qq.com)
Fangjun Chen (309647307@qq.com)
Huanling Lai (Laihl@126.com)
Weiwen Jiang (304128246@qq.com)
Linjie Zhang (1042392495@qq.com)
Ping Huang (hping331@126.com)
Hongying Cao (1171629708@qq.com)

Version: 3 Date: 06 Sep 2018

Author’s response to reviews:

Dear Gamal Badr, Ph.D.

On behalf of my co-authors, we thanks very much for the delicate work from you and reviewers to our manuscript entitled “Effects of Radix Linderae extracts on a mouse model of diabetic bladder dysfunction in later decompensated phase” (Manuscript Number: BCAM-D-18-00975). We have addressed the comments suggested by the reviewers, and the amendments were highlighted in yellow in the revised manuscript, also we provided a clean version of the revised manuscript. Point by point responses to the reviewers’ comments are listed below.
1. Ahmed Abdel Moneim (Reviewer 1):

The authors followed the comments. However, some references are not related to the work. Therefore, please remove the following references from the manuscript.


Our response: Thank you for the suggestion, and references you mentioned have been removed.

2. Mohamed Abdel-Daim, Ph.D. (Reviewer 2):

As the authors addressed the reviewers comments, I suggest acceptance of the manuscript.

Our response: We appreciated all your suggestion to accept our manuscript, and the comments were valuable to improve our manuscript.

We are sincerely grateful for your and reviewers’ valuable comments, please feel free to contact us if further revision on the manuscript is deemed necessary. We look forward to hearing from you soon.

Yours truly
Hongying Cao